Author's response to reviews

Title: Case-only study of interactions between metabolic enzymes and smoking in colorectal cancer

Authors:

- Chunhong Fan (happydou_8@hotmail.com)
- Mingjuan Jin (jinmj@zju.edu.cn)
- Kun Chen (ck@zju.edu.cn)
- Yongjing Zhang (mayun_zyi@163.com)
- Shuangshuang Zhang (qtss124@sina.com)
- Bing Liu (Liubing7362@sohu.com)

Version: 4 Date: 31 May 2007

Author's response to reviews: see over
May 31, 2007

Dear Editor,

Ms: 1390729009131154

Title: Case-only study of interactions between metabolic enzymes and smoking in colorectal cancer

Sorry for the late response. The mail was classified as junk mail at first and it takes me some times to find it out. I have revised the format of the paper according to the requirements. The detailed revisions are as following:

1. Remove the highlight in the paper.
2. Authors' contributions have changed as suggestion (on page 13).

Thank you and please let me know if further change is required.

With best regards,

Yours very sincerely,

Chunhong Fan
Department of Epidemiology and Health Statistics,
School of Public Health,
Zhejiang University